Author's response to reviews

Title: A preliminary survey of the practice patterns of United States Guild Certified Feldenkrais Practitioners(CM)

Authors:

Patricia A Buchanan (pat.buchanan@dmu.edu)

Version: 5 Date: 29 March 2010

Author's response to reviews:

I found one spacing omission on p. 6 after the first close parenthesis in the first paragraph. The addition of this space is the only change I made.

Thank you.